

Family Constitution to manage family firms' agency conflicts

Rodriguez-Garcia, P. and Menéndez-Requejo, S.

University of Oviedo

December 2020

Online at https://mpra.ub.uni-muenchen.de/120016/MPRA Paper No. 120016, posted 14 Feb 2024 14:26 UTC

FAMILY CONSTITUTION TO MANAGE FAMILY FIRMS' AGENCY CONFLICTS

Pablo Rodriguez-Garcia^a, Susana Menéndez-Requejo^b

Rodriguez-Garcia, P., & Menéndez-Requejo, S. (2023). Family constitution to manage family firms' agency conflicts. *Business Research Quarterly*, 26(2), 150-166. https://doi.org/10.1177/2340944420980444

ABSTRACT

This research examines the effectiveness of Family Constitution or Family Protocol agreements in mitigating each type of agency conflict in family firms. We performed a qualitative analysis, through a case study, and found that the succession process is the main driver for implementing this family governance mechanism. Our findings also show that a family constitution is useful in reducing three of the four agency conflicts described in the literature, specifically between family owners and managers, between family shareholders, and with the family at large. Key agreements include training and experience terms for family members to join the firm, transfer clauses of shares *inter-vivos* and *causa mortis*, and the development of family governance bodies. However, creditors are generally unaware of the protocol's existence, hindering its potential positive effects, which has important implications for practitioners. Creditors point out its potential usefulness as a hint of orderly and structured continuity of the business.

Key Words:

Family Firm, Family Constitution, Agency Theory, Case study, Protocol.

Acknowledgments

The authors gratefully acknowledge the Associate Editor, Maria José Sanchez-Bueno, and the two anonymous referees for their many insightful and useful comments that significantly improved the paper. We would also acknowledge Dr. Manuel Pavón, a prominent expert in family protocols, for sharing his valuable professional experience, as well as the helpful participation of the family and non-family managers interviewed.

This work was supported by the Spanish Science and Innovation Ministry [grant number PID2019-108503RB-I00].

^a University of Oviedo, Spain. Facultad de Economía y Empresa. Oviedo (Spain). uo91948@uniovi.es

^b *University of Oviedo, Spain.* Facultad de Economía y Empresa. Oviedo (Spain). Avda. del Cristo s/n. E-33071 Oviedo (Spain). Phone: +34 985 103 912. ORCID: 0000-0003-1641-5692. srequejo@uniovi.es

1. INTRODUCTION

Each year approximately 150,000 family firms disappear in the European Union because of an unsuccessful succession, which means an annual loss of 600,000 jobs (Official Journal of the European Union 2016). The main goal of family constitutions or family protocols is to promote the continuity of the firm (Gallo and Ward 1991). A family protocol is a family governance mechanism through which the family self-regulates the influence of their interests in the business, establishing bases to avoid or manage conflicts (Pavón 2018). The term family protocol, also known as "family constitution" or "family agreement", was settled on by Miguel Angel Gallo and John Ward (1991). A family constitution usually addresses: i) the firm's mission and values, ii) participation of the owning family in the firm, iii) ownership arrangements, iv) dividends policy and v) family involvement in boards (Montemerlo and Ward 2011). A family constitution requires a communication process between relatives in a family firm which leads to the signing of agreements about family-business relations (Gallo and Tomaselli 2006; Suess 2014; Tapies and Ceja 2011). In addition, it usually reinforces unity between family members and their commitment to the firm's success (Brenes, Madrigal, and Requena 2011; Corbetta and Montemerlo 2000). Family constitutions aim to reduce family conflicts and thus ensure the survival of the firm (Berent-Braun and Uhlaner 2012; Botero, Gómez-Betancourt, Ramírez, and López-Vergara 2015; Brenes et al. 2011). The process of discussion and writing the protocol is even more important than its content (Gallo and Tomaselli 2006; Montemerlo and Ward 2011). The implementation of a family constitution is a delicate process which can fail due to family feuds and divergence of interests (Montemerlo and Ward 2011). The family members must agree on their commitment to the business, the norms for family and business relations, and the goals of the family and the firm.

Despite the growing interest in the implementation of family constitutions in family firms (Fleischer 2018) there is scant research on this topic (Matias and Franco 2018). The pioneering descriptive study by Gallo and Ward (1991) points out that family constitutions help with firms to be professional and institutionalizes family values. Gallo and Tomaselli (2006), in a case study of twelve Spanish firms, showed that success in the implementation of a protocol depends mainly on the most influential family members' attitude and interest. A survey of 12 family firms in Costa Rica by Brenes et al. (2011) noted the family protocol as a guide for both firm and family order and structure, which improves family unity. Matias and Franco (2018), in their exploratory analysis in Portugal, also noted that family constitutions create a unity, commitment and motivational spirit crucial for the continuity of the firm. Montmerlo and Ward (2011) described the implementation process, the type of agreements and the content of a family constitution, highlighting the importance of motivation and commitment for its success. Botero et al. (2015) emphasized that agreements must be adapted to the family context and legislation in each country. Berent-Braun and Uhlaner (2012) took a sample of 64 cross-country firms from an international association of family businesses and found a positive relationship between firm performance and family governance practices (using a single scale to assess the usage of family constitution, family code of conduct, clear selection and accountability criteria, family council, formal family communication mechanisms and family meetings). Arteaga and Menéndez-Requejo (2017) analysed a sample of 530 Spanish firms and saw a positive relationship between the implementation of family constitutions and firms' performance. Although family business literature assumes that family constitutions aim to prevent or solve conflicts (Berent-Braun and Uhlaner 2012; Brenes et al. 2011; Suess 2014), we did not find any research examining which conflicts the different protocol agreements effectively mitigate, which was also pointed out by Alvarado-Alvarez, Armadans and Parada (2020).

The management of family conflicts is key to the survival of family businesses over generations (Martin 2011; Suess 2014). Agency Theory and Family Business literature allow to identify the most significant conflicts in family firms and their unique characteristics. Family firms usually have overlap between ownership and management, as well as ownership concentration, which reduces principal-agent conflicts (Anderson and Reeb 2003; Villalonga and Amit 2006), although conflicts between majority and minority shareholders or between blockholders may arise (La Porta, López de Silanes, Shleifer, and Vishny 2002; Morck, Wolfenzon and Yeung 2005; Zellweger and Kammerlander 2015). Moreover, the multiple roles of family members in the business foster singular conflicts with family members outside the business or ownership group (or "super-principals") (Chrisman, Chua, and Litz 2003; Villalonga, Amit, Trujillo, and Guzmán, 2015), while family firms' long-term orientation lessen agency conflicts with creditors (Anderson and Reeb 2003). Thus, family involvement in the firm shapes agency conflicts, which require specific governance mechanisms (Chua, Chrisman, and Bergiel 2009). Family firms need governance structures that promote cohesion, shared vision, and reduce conflict, taking into account the duality of economic and non-economic goals and the complexity of their stakeholder structures (Mustakallio, Autio, and Zahra 2002).

We aim to investigate which conflicts are reduced by the different family constitution agreements. The main research question is: How do family constitutions reduce the different conflicts in family businesses? Starting from the conflicts identified by agency theory, we investigate the usefulness of family constitutions in mitigating conflicts between managers and shareholders, between majority and minority shareholders, between "blockholders", between family shareholders and the family at large, and between shareholders and creditors. The research methodology is case study, since that allows us to examine real life events through analysing empirical evidence. Qualitative analysis emphasizes the qualities of entities and processes that cannot be measured in terms of quantity, intensity, or frequency (Denzin and

Lincoln 1994). In addition, qualitative research in family governance is underrepresented (Suess, 2014).

This study contributes to the literature in several ways. The first contribution is to enhance the understanding of how family governance mechanisms may lead to effective mitigation of conflicts in family firms (Berent-Braun and Uhlaner 2012). Building on agency theory, we investigate which agreements in family constitutions reduce each of the agency conflicts. Second, this paper contributes to expanding the literature on agency theory by considering the least studied conflict, with extended family (Villalonga et al. 2015). We jointly consider the four agency conflicts that the literature associates with family businesses. Third, this study responds to the common call for research on privately held family firms (Carney, van Essen, Gedajlovic, and Heugens 2015). Most family business research considers listed companies, although they are not the most representative. Furthermore, the control exercised by the capital market alters the importance of the different agency conflicts.

Our findings show the usefulness of family constitutions in mitigating conflicts between family owners and managers, between family shareholders, and between active family members and the extended family. Nevertheless, creditors usually ignore the existence of a family protocol, which precludes its effectiveness in reducing conflicts in this relationship. The results of this study have implications for practitioners by pointing out the most common agreements of the protocols and their effectiveness. Our research also sheds light on the reasons to implement family governance mechanisms (Suess, 2014), with succession being the main driver of family constitutions.

The remainder of the article is structured as follows: Section 2 describes the four agency conflicts and their potential relationships with family constitutions. Section 3 outlines the qualitative case study methodology, and the criteria for case selection and data collection. Section 4 presents the findings regarding the usefulness of the family constitution agreements

in mitigating each type of agency conflict. Finally, Section 5 includes the conclusions, contributions, limitations of the research and proposals for future research.

2. THEORETICAL FRAMEWORK: FAMILY CONSTITUTIONS AND AGENCY CONFLICTS

Family involvement generates additional conflicts in businesses (Harvey and Evans 1994). Moreover, the achievement of non-economic goals creates unique additional and complex challenges for family firms (Chua et al. 2009). Family governance mechanisms, primarily referring to family meetings, family councils or family constitutions, play a particularly important role in the long-term success of a family business (Mustakallio et al. 2002; Suess, 2014). Family governance mechanisms order the family's relations with the business (Mustakallio et al. 2002; Suáre and Santana-Martín 2004) and contribute to family unity and conflict resolution (Brenes et al. 2011). The family constitution may implement mechanisms for the prevention or resolution of conflicts, based on promoting communication between family members and fostering the relationship between the family and the business (Suess 2014). Grounded in agency theory, our aim is to examine how the different agreements in the family constitution can contribute to mitigate each agency conflict, that is, those between family owners and managers, between the shareholders, with the extended family, and with creditors.

2.1 Family constitutions in the manager-shareholder relationship

Information asymmetry can generate a problem of moral hazard when a family firm (principal) hires a non-family manager (agent) (Mayer 1999). The opportunism of the agent can erode the principals' interests (Barney and Hesterly 2006; Jensen and Meckling 1976). This principal-agent conflict will be least severe with a CEO of the owning family (Villalonga et al. 2015). Family management entails a natural alignment of interests with ownership (Jensen and Meckling 1976). However, firms with a family manager are not immune from this type of

agency conflict. The heterogeneity of goals between family members also threatens information flow and the firm's operation (Van den Berghe and Carchon 2003). Counterpoints to the benefits of family management include potential nepotism and altruistic behaviours that harm the firm's performance. Parents' altruism is directly related to their children's attitudes of shirking responsibilities and acting in their own interests (Buchanan 1975). Altruistic behaviour makes it difficult for owners to sanction inappropriate family practices (Schulze, Lubatkin, and Dino 2003) and even discourages the effort of family managers (Chua et al. 2009), protecting their entrenchment. This asymmetric altruism between family members (Schulze, Lubatkin, Dino, and Buchholtz 2001) limits the recruitment of talent, discourages competition between agents, reduces the professionalization of managers (Van den Berghe and Carchon 2003) and may favour overpaying family managers in relation to their effort (Chua et al. 2009).

A family constitution can promote the professionalization of family managers, by implementing succession plans, training, and the need for experience to join the firm. To avoid conflicts, the appointment and promotion of family members within the firm needs written standards in favour of competency, qualification and merit (Suess, 2014). Professionalization is a general goal of family constitutions which does not necessarily depend on the hiring of a non-family CEO (Dekker, Lybaert, Steijvers, and Depaire 2015). Associating professionalization with non-family CEOs wrongly presupposes that family members are not professionals (Bennedsen, Pérez-González, and Wolfenzon 2006; Berembeim 1990; Levinson 1971).

In the case of a non-family manager, agent-principal conflict mitigation requires designing effective incentive compensation and performance evaluation system for managers, especially in the case of family firms with both family and non-family managers (Chua et al. 2009). Career plans for non-family managers are other types of incentive that retain talent, facilitate control over the CEO, and professionalize management. In addition to agreeing on incentives, a family constitution can establish control mechanisms for the members of the board of directors and the

management team to discipline agents and correct conflicts (Zellweger and Kammerlander 2015).

2.2 Family constitutions and conflicts between shareholders

The diversity of preferences within shareholders increases the complexity of family firms (Zellweger and Kammerlander 2015). Conflicts may arise between majority shareholders ("blockholders") which can be more intense between different family branches. In turn, the risk of conflict increases as families age, since members of new generations are incorporated who have different interests and there is less social interaction between them (Gersick, Davis, Hampton, and Lansberg 1997; Mustakallio et al. 2002; Schulze, Lubatkin, and Dino 2003a), in addition to increasing the number of family members who no longer actively participate in the business (Suáre and Santana-Martín 2004). Succeeding family generations weaken familial bonds and reduce family members' identification with the firm, making family governance mechanisms more necessary (Mustakallio et al. 2002; Suess 2014).

Majority shareholders can also reinforce their power with "dual class" or pyramidal structures, acquiring voting rights that are substantially greater than they have over "cash flows". Then agency conflict may arise due to the expropriation of minority shareholders or "tunnelling" (Johnson, La Porta, López-de-Silanes, and Shleifer 2000; La Porta, López-de-Silanes, Shleifer, and Vishny 2000). Majority shareholders can take advantage of their position to obtain private benefits, in favour of their financial or non-financial interests, at the expense of minority shareholders (Grossman and Hart 1980; La Porta et al. 2000; Zellweger and Kammerlander 2015). Should investors anticipate this opportunistic behaviour from majority shareholders, it will increase the firm's cost of capital and will hinder its financing and growth (Grossman and Hart 1980; Jensen and Meckling 1976; Morck et al. 2005).

A family constitution can be useful in solving conflicts between shareholders since it promotes and establishes mechanisms of governance (Montemerlo and Ward 2011). Thus, the protocol

usually agrees rules for the inclusion of different family branches on the board of directors. In addition, disputes over transfer of ownership can be mitigated with typical family constitution agreements on testamentary limitations, marriage regimes, exit plans and contractual deals such as tag-along and drag-along. Tag-along agreements enable a shareholder to participate in the sale of their shares under the conditions agreed by another owner with a buyer. Drag-along agreements guarantee the seller the right to drag the other shareholder in a sale deal with a third party (Lacave and Gutiérrez 2010). These contractual agreements protect shareholders from opportunistic behaviours by the other owners (Lacave and Gutiérrez 2010).

Shareholder exit plans are other types of agreements that protect their interests and the continuity of the family firm by specifying sale conditions for shares in the firm. They can refer to the right of first refusal of the other partners, to the valuation of the shares, and to the establishment of a payment method in order to protect the firm from the risk of decapitalization in the event of sale. The objective of these agreements is to avoid abuse by some shareholders, in addition to neutralizing expropriation strategies, exit decisions and redistribution risks (Chung 1991). Thus, a family constitution could mitigate shareholder conflicts by means of: i) representation agreements of groups of shareholders, ii) separation between ownership and management, and iii) through ownership transfer and shareholder protection agreements.

Family firms can have an additional specific agency conflict, which has been studied less widely than the classic conflicts. Family relationships between shareholders and the other relatives who do not own or work in the firm can interfere with family business decisions (Villalonga et al. 2015). Moreover, the inheritance of family firm shares, its purchase or other form of transfer, establishes new relationships between the new shareholder and the family. The

new family shareholder represents the family, that is they act as an agent of the family. The

2.3 Family constitutions and conflicts between family shareholders and the family at large

family at large acts as a "super-principal" in this agency relationship. Their goals may be diverse and generate conflicts between shareholders and the family at large.

Passive shareholders, those who do not participate in the management or on the board of directors, may influence dividend distribution policies, limiting the liquidity and growth of the firm, against the interests of the other shareholders (Villalonga et al. 2015). Furthermore, their influence on the family members who run and control the firm can interfere in decision-making, especially in organizations with a weak governance structure. Moreover, "outsider" family members, those who are not shareholders, directors, managers or workers in the firm, can also interfere in family firm decision-making. Their family ties can allow nepotic decisions detrimental to the performance and competitiveness of the business.

The implementation of a family constitution usually entails the development of other family governance mechanisms, such as family meetings or family councils. These institutions establish areas of communication and discussion for decision-making and conflict resolution (Tapies and Ceja 2011). Family meetings or assemblies are informative, but non decision-making bodies. They usually have an informal annual meeting, partly for pleasure, and partly for communication and transmission of values. A family council is a decision-making body in which a selected representative group of family members addresses internal problems that can negatively affect the firm's competitiveness (Blumentritt, Keyt, and Astrachan 2007) and creates a nexus between the board of directors, the top management and the family (Suáre and Santan-Martín 2004; Suess, 2014). It regulates how the family operates in its relations with the firm, holds periodic meetings and institutionalizes family gatherings (Koeberle-Schmid, Kenyon-Rouvinez, and Poza 2013). The family council discusses values, behaviours and points of view about the firm (Gersick et al. 1997). In short, family governance mechanisms facilitate the transmission of family interests to the firm, contributing to the cohesion and survival of family ownership (Zahra and Pearce 1989).

2.4 Family constitutions in the relationship with creditors

Agency relationships with creditors also have their particularities in family businesses. The long term perspective of family firms, given their typically undiversified portfolio, with substantial wealth at risk, along with the goal to pass the firm on to their heirs, contributes to building trust between the firm and the financial institutions (Steijvers and Voordeckers 2009). Family firms' long-term orientation, conservative investments of family owners, and high-quality ties with financers lessen agency conflicts with creditors (Anderson and Reeb 2003; Minichilli, Brogi, and Calabró 2016). The importance of socio-emotional wealth (SEW) in relation to reputation and preserving wealth for future generations make banks perceive family firms as more creditworthy. Family firms' strong incentives for continuity in the future reduce agency conflict in the borrower-lender relationship in comparison to non-family firms (D'Aurizio, Oliviero, and Romano 2015).

In turn, family firms demonstrate great adaptation to economic contingencies, alternating between risk-aversion and risk-taking when firm continuity and economic context requires it (Minichilli, et al. 2016). Family control and the aim of passing the firm on to their heirs leads families to financially support the business in downturns (Crespí and Martín-Oliver 2015; van Essen, Strike, Carney, and Sapp 2015; Villalonga and Amit 2010). In addition, their long-term commitment allows family firms better access to debt than nonfamily firms, even in periods of financial restrictions (Crespí and Martín-Oliver 2015). Family control also influences debt maturity, facilitating access to long-term debt (Croci, Doukas, and Gonenc 2011; Díaz-Díaz, García-Teruel, and Martínez-Solano 2016). Majority family controlling shareholders will prefer long-term debt, to reduce risk-exposure, although it may generate conflicts with minority shareholders, because of the lower frequency of lender monitoring that it entails (Díaz-Díaz et al. 2016; Schmid 2013).

However, parental altruism, potential nepotism, and self-control in some family businesses could negatively influence their relationships with financial institutions (Schulze et al. 2003a; Steijvers and Voordeckers 2009). Parental altruism and nepotism may prevent external executives and protect a lack competence in family managers (Schulze et al. 2003a). Family owners and managers could also pursue personal benefits, perquisites and privileges at the expense of the business, promote special dividend payouts, excessive compensation, or riskshifting behaviour (Anderson and Reeb 2003; Claessens, Djankov, Fan, and Lang 2002; Schulze et al. 2003a). Thus, family goals may conflict with those of the banks. Soft information perceived by the loan officer in repeated personal interactions with the family firm, such as, work ethics, reputation, honesty, professionalism will be particularly important in reducing agency conflicts (Berger and Udell 2002; D'Aurizio et al. 2015; Steijvers and Voordeckers 2009). The solvency and continuity of the business are essential for the creditors. We expect family protocols to be a positive sign of the family's commitment to the business, its continuity and professionalization. The implementation of a family constitution indicates interest in implementing mechanisms for continuity of the firm, in addition to strengthening a project of unity, commitment and motivation (Gallo and Ward 1991). Furthermore, creditors may have a positive perception of non-financial aims in a family constitution. The protocol also usually addresses and therefore might mitigate potential conflicts between majority and minority shareholders. Thus, the implementation of a family constitution could improve access and financing conditions for family businesses.

3. METHODOLOGY

In order to investigate the usefulness of family constitutions in managing agency conflicts, we performed a case study. This qualitative methodology allows an in-depth analysis of current phenomena within a real context (Stake 1995; Yin 2003). The case study was carried out on family firms in Spain. Spain has a long history of institutions linked to family firms, such as the

Institute of Family Businesses (www.iefamiliar.org), one of the forerunners of the "Family Firm Institute" (www.ffi.org), and has an outstanding network of family business university chairs nationwide. Furthermore, family firms and consultants in Spain stand out for their experience in implementing family constitutions. In fact, the Spanish Senate Finance Commission recommended the formalization of family constitutions in a report in 2001.

We performed the case study through in-depth interviews, with the implementation of a family constitution as the unit of analysis. We triangulated by interviewing people in different roles in each family firm, in order to correct for inaccuracies in memory or subjective opinions (Patton 1987). The research incorporated new ideas into the study, checking that it was consistent with the initial theoretical framework and combining deductive and inductive methods.

We performed a multiple case study, thus avoiding having to rely on a single case (De Massis and Kotlar 2014), and sought to also avoid the problems from an excessive number of cases (Eisenhardt 1989). In order to determine the number of cases to study, we followed a saturation strategy, that allowed us to test the proposals and provided robustness to the research (Eisenhardt 1989; Suddaby 2006). As detailed below, the final number of cases analysed was eight (24 interviews).

3.1 Selection of cases

The case study methodology requires selection criterion that allows replication of findings in a large number of firms (Eisenhardt 1989). Although the sampling variation typical for statistical generalization was not necessary (Yin 2003), we did not analyse a random sample, but we did consider the following criteria:

- i) Geographical selection (territorial diversity).
- ii) Firm selection (different activity sectors, sizes and family generation).
- iii) Interviewee selection (different perspectives in the implementation of the family constitution).

This was not a statistical study, although we ensured diversity in the geographic selection of family firms interviewed between regions. In turn, we considered regions both large and small, based on their population with respect to the national average. Thus, we selected companies in regions of the upper population tertile (more than 5.5 million inhabitants) and in the lower tertile (up to 2.5 million inhabitants). Then in each case, we chose regions that have offered public aid to implement family constitutions and others that have not, as summarized in Table 1.

[Insert Table 1 here]

In order to identify the firms of interest, we contacted university chairs specialising in family business and regional family business associations, and consulted various sources of information, such as the press and relevant websites. The family firms interviewed must have had a family constitution for at least two years, to ensure that they have sufficient experience of them. We were looking for representative cases of private family firms. We considered private family businesses to be those with more than 50% of the firm owned by a family, as did Crespí and Martín-Oliver (2015) and Hernández-Linares, Kellermanns, López-Fernández, and Sarkar (2020) in Spain, Molly, Laveren, and Jorissen (2012) in Belgium and Luxembourg, Ang, Cole, and Lin (2000) and Steijvers and Voordeckers (2009) in US, among others. With this percentage the family has control in any case. The 50% threshold also ensures that there is only one family controlling the firm. The selection of cases also requires the second, or subsequent generations to be incorporated. Thus, we consider what Miller, Le Breton-Miller, Lester, and Cannella (2007) called true family businesses, those in which multiple family members are involved, either contemporaneously or over time. In addition, implementing a family constitution implies a certain level of business and family complexity (Suess 2014), since it produces agreements between family members involved in the firm, so it will be useful from the incorporation of new family members to the business. At the lone founder stage, the conflicts that require an agreement between potential family stakeholders have not yet arisen.

However, family governance mechanisms become more necessary as the number of family shareholders grows and their emotional cohesion and interest in the business weakens (Suess, 2014). We also selected family firms that have a goal of continuity as a family firm (Cabrera-Suárez 2005; Chua, Chrisman, and Sharma 1999). Initially, we looked for hints about that aim of continuity on the company website and in published interviews with family members. In all cases, this goal of continuity as a family firm was confirmed in the subsequent personal interview.

Table 2 summarizes the characteristics of the cases analysed with respect to the age of the firm, activity sector, size according to income and number of employees, latest generation incorporated and family branches.

[Insert Table 2 here]

We collected relevant information about each firm, consulting public sources of information, such as web pages of firms and their business environment, publications on the history of each company, as well as the Iberian Balance Analysis System (SABI) database about firms' financial statements. Subsequently, we performed three semi-structured in-depth interviews in each firm (24 interviews in total), in order to triangulate and reinforce the validity of the results. In each company we interviewed two family managers or members of the board of directors of different generations involved in the protocol (except in company E, in which only the 3rd generation was active). In each case we also interviewed a non-family member, linked to the firm as a manager, bank officer or family business consultant. Table 3 details the positions of the interviewees and the generation they represent, or their relationship as an external. The different profiles and family generations of the interviewees contributed to the robustness of the research.

[Insert Table 3 here]

3.2 Data collection and analysis

First we interviewed a family member from each company, between June and November 2019. Before each in-depth interview, we specified the topics to be addressed with the interviewee and we signed a confidentiality commitment. The validation interviews, with another family member and a non-family member, were conducted between June and September 2020. This time difference between the two rounds of interviews in each company also allowed us to delve into issues which emerged in the first interviews. The interviews investigated the content and usefulness of the family constitution for managing conflicts in the family firm. The questions of interest were organized into five major topics:

- i) Family constitution: reasons for implementation, process, most important content, characteristics.
- ii) Professionalization (family and non-family managers): requirements for family members to join the firm.
- iii) Shareholder agreements: deals and limitations on ownership transfers, blockholders and minority shareholder protection agreements.
- iv) Relations with creditors: mission, values, shareholders' commitment, long-term goals.
- v) Corporate and family governance agreements: usefulness of the board of directors and the family council (characteristics, performance and utility).

Each interview lasted between 1 and 1.5 hours. We carried out half of the interviews by telephone and the other half in person. We addressed emerging topics during the interviews in order to delve into the specific circumstances of the implementation of the family constitution. We recorded most of the interviews and transcribed them within six hours of completing them. Subsequently we reviewed the content of each interview to check the consistency of the interpretation. Likewise, we sent a draft of the article to the interviewees, to check the correct

interpretation of their contributions. The aim of the research is the generalization of a theoretical framework (analytical generalization) and not the mere enumeration of frequencies of a sample (statistical generalization). The purpose of the research is to develop a logic that can be transferred to other cases (Yin 2003) about the usefulness and content of a family constitution. We also investigated firms' ownership and corporate governance structures, as summarized in Table 4. In five of the firms, 100% of the ownership belonged to the family, and it was over 60% in the remaining three cases (two of these also had some minority family shareholders). The board of directors had independent directors in all firms that had some non-family owners, but only in one of the 100% family-owned firms. Seven out of the eight companies had a family CEO. Only one firm had CEO duality.

[Insert Table 4 here]

In the following section we show contextual descriptions and quotes from the interviewees to examine the usefulness of the agreements included in the protocols. The information was simplified before being organized, producing tables and texts that make it easier to analyse the collected data. We also examined the circumstances of the family firm when it decided to develop the family constitution, the family generation that promoted it, and whether there were problems of professionalization or family conflicts before and during the discussion of the protocol.

4. RESULTS AND DISCUSSION

Analysis of the in-depth interviews provided important information about factors that encourage the implementation of a family constitution, and the usefulness of the agreements in managing agency conflicts. All of the family constitutions examined included clauses on the professionalization of the family firm, the protection of shareholders, the continuity of the firm and the development of family governance and corporate governance bodies.

4.1 Circumstances that promote the implementation of a family constitution

Despite the growing amount of research devoted to family governance, there is no consensus as to why family firms decide to implement family governance mechanisms (Suess, 2014). The interviews we performed allowed us to identify the main reasons and circumstances that motivated implementation of family constitutions. We asked about who promoted it, which generation it was implemented in, and what the reasons were for deciding to develop a family constitution. We also asked about the development process and how long the signed document was.

The main driver for implementing a family constitution was succession in almost all of the cases examined, in particular, the first succession process with more than one family candidate. The generational change in family businesses often causes a level of stress in the family and in the company that affects the dynamics of both, generating conflicts, lack of competitiveness and loss of resources (Pavón 2018). Table 5 shows interviewees' observations about the reasons for starting the process, when they chose to do it, and the aims. Most firms decided to implement the protocol immediately after a succession process due to organizational problems and fear of decision-making that emerged. Successors pointed out problems of a lack of professionalism, hierarchical disorder, and difficulties in decision making. Only three companies implemented family constitutions prior to succession. In two of those three companies, the interviewees said that the main reason for implementing the family protocol was to facilitate a smooth, professional transition.

"It was necessary to resolve a situation of hierarchical confusion where everyone is dedicated to everything. My father was still making decisions, me too, so there were three different opinions on the same issue" (D1).

The initiative to implement a family constitution started from the second generation in most of the firms examined. In only two cases were the founders the promoters, and in only one was a relative not involved in the management of the firm. When the third or subsequent generation took the initiative, the complexity of those firms was similar to that of a second generation firm, because the shareholders were the only children, explaining the absence of conflicts between heirs.

[Insert Table 5 here]

The discussion of a family constitution can generate conflicts that threaten the continuity of the firm. In three of the cases, the process of drafting the protocol gave rise to family tensions. In two cases the conflict ended with the sale of shares by a family branch and ownership restructuration. Although interviewees indicated that developing a family protocol is demanding, all of them agreed that the effort was worth it.

"Significant effort for two years, without knowing very well what a family constitution was, with doubts about its usefulness and with meetings on Saturdays, sometimes until eleven at night. Without a doubt, the result is worth it" (A1).

"The writing of the protocol generated differences in the family" (B1).

"The process of implementation was delayed due to conflicts between family branches. Finally, a family branch acquired all the shares and the protocol drafting process was restarted" (A1).

"The implementation of the protocol generated tensions in the family and also in the family firm" (C1).

It is interesting to note the importance of a consultant in Firm B to remove differences and motivate family members into participating in the writing of the family protocol:

"Another consultant was contacted who got the interest and support of the most reluctant family members to implement the protocol" (B1).

Although the process of the family constitution has a risk of failure that can affect family relationships inside and outside the business (Montemerlo and Ward 2011), in two cases the protocol discussion process resolved conflicts before implementation.

"During the discussion of the protocol, a conflict between one of the siblings and an in-law who participated in the management of the firm was resolved" (G1).

"The development of the protocol allowed the resolution of a conflict between a descendant and the rest of the family" (H1).

Table 6 summarizes the data related to the drafting of the document in the firms examined, time taken to draft it, number of family owners and signatories, length, versions, and age of the family constitution. The average family constitution needed twelve months for implementation. The average length was 52 pages, with the shortest being 15 pages long and the longest 108 pages. The numbers of signatories varied considerably, averaging 9 but ranging between 4 and 21. Although the signing of the family constitution is voluntary, all family owners have signed it, except for company B, which has 58 family shareholders (out of a total of 143 owners). The high number of owners in this case meant that only representatives of each family group signed the protocol, 12 in total. However, at present this company has initiated a process for all family owners to sign the protocol. The decision to sign the family constitution shows the interest and motivation to be part of the agreements adopted, despite the difficulties and effort required for its implementation. In some companies, in addition to family shareholders the protocol had been signed by non-owner family members of legal age, including some in-laws and members of the previous generation who had already transferred ownership to their successors.

[Insert Table 6 here]

Regarding the content of the agreements, the most important issues addressed by family constitution in the firms interviewed can be grouped into:

- i) Agreements on criteria for family participation in the management teams.
- ii) Agreements regarding the transfer of ownership.
- iii) Development of family and corporate governance bodies.

The interviewees concurred, highlighting these three topics as the most important in family constitution. Next, we look more deeply at the content of these agreements, examining their usefulness for managing conflicts of interest and agency problems in family firms.

4.2 Usefulness of family constitution in reducing conflicts with managers

The incorporation of training and experience requirements for family members who want to participate in the management teams limits owners' nepotism in the selection of managers and promotes professionalization of the firm. These agreements are especially important in firms with a family successor, since the altruistic behaviours of parents can encourage nepotism and a lack of professionalization (Caselli and Gennaioli 2013).

The interviewees agreed that the implementation of a family constitution promoted the professionalization of the firm. All of the family protocols examined included training and experience requirements for family members to participate in the firm's management teams. In some cases the family constitutions detailed the minimum age for inclusion and even a required level of a second language.

"Minimum age 25, degree and masters qualifications. Experience between two and five years in another firm (preferably in a related sector). Second language advanced level and third language medium level. Incorporation into a rotating training program" (H1).

The family members interviewed agreed that they would not lower the requirements. They all preferred a family successor, but if there is no one who meets the required training criteria, they will agree to hire an external manager.

"The protocol establishes that family members who work in the firm have to demonstrate professional performance to continue in the family business" (B1).

"In the next ten years we will have to incorporate a third-generation family member into the management, God willing!, but it is not essential" (D1).

"We deem that it is important to choose well, by meritocracy, and that it will not be a problem if in the next generation we have to resort to external professionals" (E1).

Regarding the incorporation of in-laws into the firm, most family constitutions specifically prohibit their participation in management teams. Most of the interviewees did not consider inlaws as part of the family firm. They agreed that the participation of in-laws in firms increased the risk of conflicts in the family. Their participation is only allowed in three cases, if they meet certain training and experience requirements.

"It would be very difficult for me to argue with a family member and go to eat at his/her house on Sunday. I think limiting access for in-laws is very good for preserving family life" (F1).

"The in-laws are not Relatives. They can access the firm, but under the same conditions as an outsider" (B1).

"In-laws may work in our group of companies as long as they meet the following requirements: that their job position is designed, that their functions are specified, that they have appropriate training for the position and that they are monitored by external directors" (D1).

In short, the family constitution establishes procedures that promote the professionalization of the firm.

"Of course, the family constitution promotes the professionalization of the firm due to the demands of training and experience" (A1).

"The protocol favours professionalization because it establishes management systems that, although very common in large companies, are not so much in family businesses" (C3).

"Yes, the protocol promotes the professionalization of the firm through the development of standards and the strategic plan" (G1).

Non-family managers we interviewed stated that family constitution clarified the career plans of family and non-family members. They also felt that the demanding requirements agreed in the protocol for positions of greater responsibility and remuneration were a great professional

incentive for them, because it limited access to managerial positions for family members who did not meet those demands. The usual business continuity agreements were another motivation for non-family managers to align their goals with those of the business.

"The training and experience requirements agreed in the family protocol guarantee that the positions are not decided only by the family relationship" (D3).

"The existence of a family protocol also gives us non-relatives a future perspective. Just as the continuity objective gives banks confidence, it also gives security to non-family members" (C3). According to the above, we propose that the family constitution improves the professionalization of the firm and limits nepotism, positively influencing the resolution of agency problems between managers and shareholders:

Proposition 1.a: Family constitutions promote the professionalization of the firm.

Proposition 1.b: Family constitutions reduce conflicts between shareholders and managers.

4.3 Usefulness of family constitution in reducing conflicts between shareholders

The interviewees highlighted the importance of family constitutions in regulating the relationship between shareholders. The family constitution develops agreements to avoid conflicts between "blockholders" in different branches of the family and contains clauses for the transfer of ownership, in addition to protecting minority shareholders.

The family constitution includes clauses of representation and defence of the shareholders' interests. In most companies, each family branch has a similar share in the board of directors. Owners usually discuss their concerns informally in their family branch, whose representatives forward the proposals to the appropriate governing bodies. In some cases, the protocol establishes a minimum percentage of shares of stock to constitute a new group of family members.

"Each owner is represented and protected by his/her family group" (B1).

In most firms, family constitutions agreed limitations on ownership transfer, through clauses related to matrimonial property regimes, and *inter-vivos* or *causa mortis* ownership transfers. The objective of these agreements was to retain family ownership. Although the relatives were free to sell their shares, the constitutions established agreements in order for property to remain in the family.

"We do not want the family firm to be a mousetrap that cannot be left" (E1).

The most common agreements referred to the right of first refusal by the rest of the family, although sometimes they began with the seller's family branch and continued with the other family branches. Share valuation agreements were also frequent in the event of an "exit". The agreed value per share is what the firm would pay the owner who decided to leave the firm. This valuation was to be reviewed annually and did not correspond in any case with the market value of the shares, but was agreed by the owners, as were the payment terms. The firms guaranteed the purchase of the shares under the agreed conditions, therefore they did not establish liquidity premiums. The aim of these agreements is to retain ownership control and avoid decapitalization of the firm if any partners depart. Two of the family constitutions included tag-along and drag-along anti-lock clauses. Tag-along agreements allow minority shareholders to sell under the conditions agreed by another owner. Drag-along agreements protect majority shareholders' interests, as they guarantee the right to drag other shareholders after a sale agreement to a third party.

The family constitutions examined established anti-block mechanisms to avoid paralysis of the governance bodies and abuse by some of the parties. Blockades occur when it is not possible to achieve the necessary majorities to reach agreements in the normal course of events. In all cases they established anti-blocking mechanisms through qualified majorities, and in cases where there were only two family branches, through the casting vote of the president. In one of the family protocols with two family branches, they agreed that to avoid blockages, one of the

branches would have 51% of the ownership and this dominance position would pass to the other family branch in the subsequent generation.

"We have defined how to resolve blocking situations through mediation and arbitration. Now that we get along very well, we have decided what to do when we would get on badly. We have also agreed on a penalty system" (C2).

"So far there has been no problem of blocking decision-making, but if that were the case, we have agreed in the family constitution that the owner of the 1% could unlock the agreement" (D2).

"Family protocol establishes that decision making has to be by consensus. If consensus is not reached we are obliged to try again. If it is finally impossible, the President will decide, who has a casting vote" (F1).

Companies with few owners do not have minority shareholders, but ownership is distributed equally. The firms with some minority shareholders indicate that there are no conflicts between majority and minority owners, since they can approach the representatives of their family branch if they believe that their rights are being violated. Only one of the interviewees indicated that the dividend distribution could be an annual source of discrepancies with minority shareholders. But, transparency and information about the goals and growth plans of the business facilitated understanding between all shareholders.

Shareholder protection agreements and their importance in all of the cases examined are consistent with the following proposals with respect to the usefulness of family constitutions in reducing agency conflicts between shareholders:

Proposition 2.a: Family constitutions reduce conflicts between blockholders.

Proposition 2.b: Family constitutions reduce conflicts between majority and minority shareholders.

4.4 Usefulness of family constitutions in managing conflicts between family shareholders and the family at large

Family constitutions usually agree on other family governance mechanisms, which encourage decision-making and communication between family members. Family governance generates areas for discussion and resolution of family conflicts and limits interference in decision-making.

Before the implementation of the family constitution, none of the firms had established a family council. The interviewees valued its usefulness in promoting communication, cohesion and avoiding interference in the firm. One of the interviewees summarized the usefulness of the family council saying:

"The greatest utility of our family council, which we have always encouraged, is creating, maintaining, and promoting family-firm and firm-family cohesion" (A1).

"The family council has improved communication. We aim that neither minority, in-laws nor passive owners can significantly influence the firm's decisions" (E1).

"A family constitution does not favour communication, but a family council does. There is undoubtedly a risk of interference in the management of the firm, but the family constitution limits this interference" (G1).

Once family councils were established, their agreements are given particular importance.

"Decisions will depend on the family council at the request of the board of directors" (F1).

"One of the aims of the family constitution is to improve communication. The family council allows us to know the concerns of the family members. However, it cannot be denied that there may be attempts to influence decision-making" (B1).

The family meeting or family assembly is another family governing body in which the entire family participates. Its mission is to report on the progress of the firm, for the family to get to know each other and transmit the values and contents of the family constitution. The family

assembly is generally held in the firm, to provide space for the whole family to meet and to show the facilities to the younger generations. Some interviewees pointed out that their meetings take place in the family home. There were family assemblies in five of the eight companies examined. In the other three, the interviewees said that they were still firms with few family members, so the members of family assemblies would be almost the same as in the family councils. Family assemblies are very useful for promoting unity, communication and interest in the firm's activities among family members, especially the youngest. In one case, they even created a junior committee with activities linked to a small budget, to promote interest in the firm for new generations. Family cohesion and the transmission of values are extremely useful for aligning interests, reducing conflicts and facilitating generational change. In the family assembly they propose topics for debate in the family council. It is an opportunity for all family members to express their ideas, both those who work in the firm and those who do not.

The family protocol may also contain agreements regarding the relationship with the management team and the board of directors, to avoid interference.

"If you don't work in the business, you can't make decisions. You have to delegate your vote to a relative who works in the firm" (C2).

"The main contribution of our protocol to avoid tension and interference in management is the agreement not to hire in-laws" (F1).

Thus, we see that the development of family governance bodies promotes organization, communication, the transmission of values and decision-making. Although the attempt to interfere may be inevitable, the family council and the family meeting are the main frameworks for communication and transmission of values in the family firm. These family governance mechanisms limit the scope of possible interference and reduce conflicts between family members. Based on the previous arguments, we make the following propositions:

Proposition 3.a: Family constitutions favour communication in the family.

Proposition 3.b: A family constitution reduces the interference of relatives from outside the firm.

4.5 Usefulness of family constitutions in reducing conflicts with creditors

The protocol is a private family document. In Spain, the content of the family constitution or its existence can be published in the Commercial Register (Royal Decree 171/2007), but companies rarely do so. The interviewees stated that they did not usually inform the banks either. In only two of the cases we examined (Firms B and C) did the creditors know that the family had implemented a protocol. The ignorance on the part of the creditors prevents them from assessing the agreements included in the family constitution.

"Creditors don't know that we have a family protocol. It is something internal" (A2).

"No one has asked us about the protocol" (F1).

Although bank creditors sometimes hold a seat on the board of directors (Fama, 1985; Kaplan and Minton, 1994), this was not the case in any of the eight firms studied. Nor have these companies received proposals from the banks to participate in ownership. The two banking professionals interviewed indicated that:

"The involvement of the bank in the ownership and on the board of directors can generate a conflict of interest, in addition to limiting the financing to a single creditor" (A3).

"As a general rule, I consider it an error for the bank to participate in the ownership or on the board of directors of the firm, although obviously each specific case must be analysed" (B3). Almost all of the firms had long-term multi-banking relationships with between 2 and 6 banks simultaneously, with gradual incorporation of new financial entities. Only one of the companies was financed by a single bank. In all cases, the main creditor belonged to the top five Spanish banking groups. The level of indebtedness of these companies was between 25% and 50%. Family members interviewed highlighted the personal frequent interaction with bank officers.

"The banks we work with know the firm and its managers very well. That gives them a guarantee and that there is continuity from parents to children" (D2).

"We usually work with the same banks. They know the firm and the managers very well, but they do an independent borrower risk assessment of each operation" (D2).

All the interviewees agreed that the family status of the firm encouraged agreements with creditors. Family values are expected to be aligned with the interests of creditors, which improves access to financing and its conditions.

"Lately, it is highly valued to be a family business. They consider us strong, long-term organizations. Financial institutions are more aware that the family business has an added value" (D1).

"Being a family firm positively affects the relationship with creditors. Most family firms have rooted principles, seriousness and a will for continuity. We have been working with some suppliers for forty years" (F1).

Creditors pointed out that banks were always interested in learning new information about companies and their managers, decision-making criteria, family situations, training, experience, or the proximity of a generational transition. They also appreciated firms informing the bank of the existence of a family constitution, as it is a sign of continuity of the business. This encourages them to grant loans, although it does not influence the financing conditions.

"The risks assumed by financial institutions must be based on trust and transparency between the bank and the client. Companies should communicate and value the existence of the protocol" (B3).

"The protocol is useful information for the bank, favourable for granting financing, although not for financing conditions. It also reduces uncertainties about the succession" (A3).

"Undoubtedly, one of the aspects that concern financial institutions in long-term risk analysis is the succession of the business, especially in family firms. The existence of a family constitution favours the continuity of business in an orderly and structured way" (B3).

Therefore, the case analysis we performed does not allow us to conclude that a family constitution improves the agency relationship with creditors. However, the information provided by the interviewees has important practical implications, as we detail in the concluding section. In brief, knowing that the firm has implemented a family constitution would convey positive information about business continuity to creditors.

Proposition 4: The general ignorance of the existence of a family constitution means that creditors are unable to positively evaluate its content.

5. CONCLUSIONS

The unique characteristics of family firms warrant specific mechanisms for family governance, which can make a difference to their success (Mustakallio et al. 2002). The supervision and monitoring role of boards of directors is not enough. Family governance mechanisms, primarily referring to family meetings, family councils, and family constitutions order the family's relations with the business and contribute to family unity and conflict resolution (Mustakallio et al. 2002; Suáre and Santana-Martín 2004; Suess, 2014). Family constitutions are the least researched of the three family governance mechanisms (Suess, 2014). We have to bear in mind that reliable information on family protocols is extremely difficult to obtain, since it is private and confidential information of the family owners. Starting from agency theory and previous literature pointing out that family constitution aims to prevent or solve conflicts (Berent-Braun and Uhlaner 2012; Brenes et al. 2011; Suess 2014), this research examines which conflicts are effectively mitigated by the different protocol agreements.

We performed a qualitative analysis using a case study methodology, in order to focus our attention within family firms. The research identified succession as the main trigger for the development of a family constitution. The interviewees agreed that succession revealed organizational problems in the management of the firm related to a lack of planning, hierarchical disorder and fear of decision-making. This explains why some of the most important content in the family constitutions are the professionalization of the firm and the development of family governance bodies.

Our research confirms family constitutions as a valuable mechanism for family governance, in particular to manage conflicts with the managers, between shareholders and with the extended family. The most intriguing finding is that creditors are generally unaware of the protocol's existence, hindering its potential positive effects. However, the banking professionals interviewed highly value that family firms implement a family constitution. The family protocol signals the aim of continuity of the business in an orderly and structured way, which is of great interest to the creditors. In addition, the protocol favours the professionalization of management, since it establishes requirements for family members to join the firm. This is a key issue for the bank in credit risk analysis, along with a successful generational transition. What is relevant for the bank is not whether or not the firm is family-owned, but whether it has professionalized management and continuity.

Our study shows that family constitutions usually reach agreements on training and experience requirements for managers, on ownership transfers, and on the development of other family governance mechanisms. The interviewees stated that the agreements on training and experience for family members to join the firm promote professionalization, limit owners' nepotism and reduce conflicts between shareholders and managers. The family members also noted that family constitutions develop agreements on ownership transfers, which mitigate conflicts related to firm exits, inheritance and the participation of in-laws in ownership. The agreements include drag-along and tag-along clauses, *inter-vivos* and *causa mortis* transfer of share agreements, share valuation and payment method if a family member or group requests

to sell their shares, and exit agreements with right of first refusal clauses. Likewise, the interviews revealed key issues in family constitutions that prevent extended family interference, in terms of the development of other family governance mechanisms, such as family councils and family meetings, which open areas of communication, exchange of opinions and transmission of values, promoting family cohesion and the continuity of the firm.

5.1. Contributions

This research contributes both to the literature on family governance and to agency theory. Firstly, we contribute to the literature on family governance, by investigating how the agreements on family constitutions mitigate the main agency conflicts, that is, conflicts of family owners with managers, between family shareholders, with the extended family and with creditors, taking into account the unique nature of family ownership. Our findings broaden the understanding of the least studied of the three main family governance mechanisms (Suess, 2014), which as we saw, promotes the other two mechanisms in turn; family councils and family meetings. The qualitative analysis, which is also underrepresented in research into family governance mechanisms (Suess, 2014), allowed us to deepen our knowledge of the characteristics of the protocol and its effectiveness in reducing conflicts in family firms. Our findings also shed light on the reasons for implementing family governance mechanisms (Suess, 2014), with succession being the main driver of the family constitution.

Secondly, we contribute to the incipient study of the agency problem related to relationships between family shareholders and the family at large (Villalonga et al. 2015). This is an additional conflict specific to family firms, relating to interference in the business decisions of passive shareholders and extended family. The implementation of the protocol helps mitigate this type of conflict, by promoting information and communication between family members and channelling their demands. We also examined the usefulness of the family constitution as a tool to mitigate other agency conflicts in family firms. While the family protocol achieves to

mitigate conflicts among family shareholders, with the extended family and with managers, the same does not occur in their relationship with creditors. Family firms often do not inform creditors about their family constitutions. However, creditors point out its potential usefulness as a hint of orderly and structured continuity of the business.

Thirdly, we respond to the increasing calls for deeper research on private family firms (Carney et al. 2015). Although most family firms are privately held, studies on listed family firms predominate. Private family businesses have largely been underrepresented in research because of the limited availability of data. But agency conflicts for listed firms have their peculiarities, due to the control exercised by the stock markets. Therefore, the conclusions drawn from listed family firms may differ from those derived from the study of privately held family firms. Likewise, qualitative research on family governance is scarce (Suess, 2014). Our research contributes by providing new qualitative evidence on family governance practices in private family firms.

5.2. Practical implications

Our findings have several practical implications for family owners and practitioners, summed up in the usefulness of implementing a family constitution and communicating its existence to creditors. Family business leaders and owners should promote appropriate family governance systems. The results of our qualitative research advise family firms to develop family protocols to favour their professionalization, minimize conflicts between shareholders and avoid family interference in the business. Key agreements include training and experience terms for family members to join the firm, transfer clauses of shares *inter-vivos* and *causa mortis*, and the development of other family governance mechanisms, such as family councils and family meetings. Our research finds a great shortcoming in the disclosure of family protocols to creditors. A family constitution could improve the firm's relationship with creditors but this does not happen because creditors are not aware of it. The banking professionals we interviewed

indicated that it should be the companies that communicate and stress the value of implementing the protocol. It makes the business's continuity more credible, improving the perception of its creditworthiness. They also recommend developing the legal regulations, in relation to the commercial register, to be able to reliably incorporate this information. This would be very helpful. Our findings also speak to loan officers, pointing to the usefulness of inquiring about the existence of a family constitution in their personal interactions with managers. Agreeing on a family protocol generally shows a stronger commitment to the continuity of the family business, which is relevant soft information for assessing borrower risk. They usually incorporate succession plans and agreements on shares sale and financing. These agreements between shareholders in the protocols are especially useful to ensure their commitment in downturns. Our results also have implications for policymakers in designing the mechanisms to support the continuity of family businesses, which are key for national economies. Policies aimed at granting financial aid to implement protocols will be of interest. Likewise, it is advisable to promote knowledge of family constitutions and their characteristics by family owners and managers, business consultants, creditors and business associations.

In Spain there is an interesting regulation in favour of the disclosure of family protocols, though it is little known or used. The Royal Decree (Act) 171/2007 (February 9, 2007) regulates the voluntary publication of family constitutions. This law establishes three levels of publication of family protocols through the Commercial Register, namely: a) mention that the firm has a family constitution, b) register all or part of the protocol with the annual financial statements, or c) public registration of some of its agreements. This experience may inspire the design of similar regulations in other countries. We also encourage policymakers to spread the knowledge of these tools to companies, consultants and creditors.

5.3. Future research

Despite the legal support, the lack of disclosure regarding the existence of a family protocol by its signatories leads us to pose new research questions, which future research might address. Does the goal of socio-emotional wealth (SEW) preservation make it preferable not to report on family constitution? Concern for preserving family control of the firm is a critical element of family SEW (Gómez-Mejía, Cruz, Berrone, and De Castro 2011; Gómez-Mejía, Haynes, Núñez-Nickel, Jacobson, and Moyano-Fuentes 2007), which may advise against disclosing protocol information, considering it private. On the other hand, we also wonder whether the implementation of a protocol contributes to preserving SEW (Gómez-Mejía, Makri, and Larraza 2010) or whether some protocol agreements (such as those making it difficult for family members to achieve managerial positions) erode SEW (Gómez-Mejía et al. 2010; Kellermanns, Eddleston, and Zellweger 2012). Which SEW components, in the "FIBER" sense (Berrone, Cruz, and Gómez-Mejía 2012), enhance or erode the family constitution: family control and influence (F), identification of family members with the firm (I), binding social ties (B), emotional attachment of family members (E) and renewal of family bonds to the firm through dynastic succession (R)?

5.4. Limitations

Our research is not without limitations. These include the limitations of case study methodology, the selection of cases, and the interpretation of the data by the researcher. Moreover, Spanish firms may have a specific culture, which may influence the findings. In any case, the nature of case study methodology prevents extrapolation of the results. Although Spain is a particularly suitable setting for analysing family constitutions, future studies could consider extending this research to other institutional environments. Another extension of our research would be to analyse whether the conclusions hold in firms where the family owns less than 50%, or the case of firms with several owning families, each of which could have their own

family protocols. Likewise, future studies could provide statistical analysis in those aspects that may be quantifiable.

Above all, we hope to open a discussion about the usefulness and characteristics of family constitutions in promoting the continuity of family firms.

6. REFERENCES

- Alvarado-Alvarez, C., Armadans, I., Parada, M. J., 2020. Tracing the Roots of Constructive Conflict Management in Family Firms. Negotiation and Conflict Management Research, 13(2), 105-126.
- Ang, J.S., Cole, R.A., Lin, J.W., 2000. Agency Costs and Ownership Structure. Journal of Finance, 55, 81-106.
- Anderson, R.C., Reeb, D.M., 2003. Founding-Family Ownership, Corporate Diversification. and Firm Leverage, Journal of Law and Economics, 46, 653-684.
- Arteaga, R., Menéndez-Requejo, S., 2017. Family Constitution and Business Performance: Moderating Factors. Family Business Review, 30(4), 320-338.
- Barney, J.B., Hesterly, W., 2006. Organizational Economic: Understanding the Relationship between Organizations and Economic Analysis. The SAGE Handbook of Organization Studies, 111-148.
- Bennedsen, M., Pérez-González, F., Wolfenzon, D., 2006. Do CEOs matter. NYU Working Paper, No. FIN-06-032.
- Berger, A.N., Udell, G.F., 2002. Small business credit availability and relationship lending: the importance of bank organizational structure. The Economic Journal, 112(477), 32-53.
- Berent-Braun, M.M., Uhlaner, L.M., 2012. Family governance practices and teambuilding: paradox of the enterprising family. Small Business Economics, 38(1), 103-119.
- Berrone, P., Cruz, C., Gómez-Mejía, L. R., 2012. Socioemotional Wealth in Family Firms: Theoretical Dimensions, Assessment Approaches, and Agenda for Future Research. Family Business Review, 25(3), 258-279.
- Blumentritt, T., Keyt, A., Astrachan, J., 2007. Creating an Environment for Successful Nonfamily CEOs: An Exploratory Study of Good Principals. Family Business Review, 20(4), 321-335.
- Botero, I., Gómez-Betancourt, G., Ramírez, J., López-Vergara, M., 2015. Family protocols as governance tools. Journal of Family Business Management, 5(2), 218-237.
- Brenes, E.R., Madrigal, K., Requena, B., 2011. Corporate governance and family business performance. Journal of Business Research, 64(3), 280-285.
- Buchanan, J.M., 1975. The Samaritan's Dilemma. Altruism, morality and economic theory. Edmund Phelps. New York: Russell Sage Foundation, 71-75.
- Cabrera-Suárez, K., 2005. Leadership transfer and the successor's development in the family firm. The Leadership Quarterly, 16(1), 71-96.

- Carney, M., van Essen, M., Gedajlovic, E.R., Heugens, P., 2015. What do we know about private family firms? A meta-analytical review. Entrepreneurship: Theory and Practice, 39(3), 513-544.
- Caselli, F., Gennaioli, N., 2013. Dynastic Management. Economic Inquiry, 51(1), 971-996.
- Chrisman, J. J., Chua, J. H., Litz, R., 2003. A unified systems perspective of family firm performance: An extension and integration. Journal of Business Venturing, 18(4), 467-472.
- Chua, J.H., Chrisman, J.J., Sharma, P., 1999. Defining the family business by behavior. Entrepreneurship theory and practice, 23(4), 19-39.
- Chua, J.H., Chrisman, J.J., Bergiel, E.B., 2009. An agency theoretic analysis of the professionalized family firm. Entrepreneurship Theory and Practice, 33, 355-372.
- Chung, T.Y., 1991. Incomplete Contracts, Specific Investments, and Risk Sharing. The Review of Economic Studies, 58(5), 1031-1042.
- Claessens, S., Djankov, S., Fan, J.P.H., Lang, L.H.P., 2002. Disentangling the Incentive and Entrenchment Effects of Large Shareholdings. Journal of Finance, 57(6), 2741-2771.
- Corbetta, G., Montemerlo, D., 2000. Family Protocols: Our Experience in Italy. The Family Business Network Newsletter, 26, May.
- Crespí, R., Martín-Oliver, A., 2015. Do Family Firms have Better Access to External Finance during Crises? Corporate Governance: An International Review, 23, 249-265.
- Croci, E., Doukas, J.A., Gonenc, H., 2011. Family control and financing decisions. European Financial Management, 17(5), 860-897.
- D'Aurizio, L., Oliviero, T., Romano, L., 2015. Family firms, soft information and bank lending in a financial crisis. Journal of Corporate Finance, 33, 279-292.
- Dekker, J., Lybaert, N., Steijvers, T., Depaire, B., 2015. The effect of family business professionalization as a multidimensional construct on firm performance. Journal of Small Business Management, 53(2), 516-538.
- De Massis, A., Kotlar, J., 2014. The case study method in family business research: Guidelines for qualitative scholarship. Journal of Family Business Strategy, 5(1), 15-29.
- Denzin, N. K., Lincoln, Y. S., 1994. Handbook of qualitative research. Thousand Oaks, CA: SAGE.
- Díaz-Díaz, N., García-Teruel, P., Martínez-Solano, P., 2016. Debt maturity structure in private firms: Does the family control matter? Journal of Corporate Finance, 37, 393-411.
- Eisenhardt, K., 1989. Building Theories from Case Study Research. Academy of Management Review, 14(4), 532-550.
- Fama, E.F., 1985. What's different about banks? Journal of Monetary Economics, 15(1), 29-39.
- Fleischer, H., 2018. Family Firms and Family Constitutions: A Legal Primer. European Company Law, 15(1), 11-20.
- Gallo, M.A., Tomaselli, S., 2006. Formulating, implementing and maintaining family protocols. In P.Z. Poutziouris, K. X. Smyrnios, S. B. Klein (Eds.), Handbook of research on family business (pp. 298-316). Northampton, MA: Edward Elgar.

- Gallo, M.A., Ward, J., 1991. Family Protocol. Barcelona, Spain: Technical Note Research Division, IESE, DGN-448.
- Gersick, K.E., Davis, J.A., Hampton, M.M., Lansberg, I., 1997. Generation to Generation: Lifecycles of the Family Business. Boston: Harvard Business School Press.
- Gómez-Mejía, L., Cruz, C., Berrone, P., De Castro, J., 2011. The bind that ties: socioemotional wealth preservation in family firms. Academy of Management Annals, 5(1), 653-707.
- Gómez-Mejía, L.R., Haynes, K.T., Nuñez-Nickel, M., Jacobson, K.J.L., Moyano-Fuentes, J., 2007. Socioemotional wealth and business risks in family-controlled firms: Evidence from Spanish olive oil mills. Administrative Science Quarterly, 52(1), 106-137.
- Gómez-Mejía, L.R., Makri, M., Larraza, M., 2010. Diversification Decisions in Family-Controlled Firms. Journal of Management Studies, 47 (2), 223-252.
- Grossman, S. J., Hart, O. D., 1980. Takeover bids, the free-rider problem, and the theory of the corporation. The Bell Journal of Economics, 11(1), 42-64.
- Harvey, M., Evans, R.E., 1994. Family business and multiple levels of conflict. Family Business Review, 7(4), 331-348.
- Hernández-Linares, R., Kellermanns, F.W., López-Fernández, M.C., Sarkar, S. 2020. The effect of socioemotional wealth on the relationship between entrepreneurial orientation and family business performance. Business Research Quarterly, 23(3), 174-192.
- Jensen, M.C., Meckling, W.H., 1976. Theory of firm: Managerial behavior, agency costs and ownership structure. Journal of Financial Economics, 3(4), 305-360.
- Johnson, S., La Porta, R., López-de-Silanes, F., Shleifer, A., 2000. Tunneling. American Economic Review, 90, 22-27.
- Kaplan, S.N., Minton, B.A., 1994. Appointments of outsiders to Japanese boards: Determinants and implications for managers. Journal of Financial Economics, 36(2), 225-258.
- Kellermanns, F.W., Eddleston, K.A., Zellweger, T.M., 2012. Extending the socioemotional wealth perspective: A look at the dark side. Entrepreneurship Theory and Practice, 36, 1175–1182.
- Koeberle-Schmid, A., Kenyon-Rouvinez, D., Poza, E., 2013. Governance in Family Enterprises: Maximising Economic and Emotional Success. Springer.
- Lacave, I.S., Gutiérrez, N.B., 2010. Specific investments, opportunism and corporate contracts: A theory of tag-along and drag-along clauses. European Business Organization Law Review (EBOR), 11(3), 423-458.
- La Porta, R., López-de-Silanes, F., Shleifer, A., Vishny, R., 2000. Investor protection and corporate governance. Journal of Financial Economics, 58(1-2), 3-27.
- La Porta, R., López de Silanes, F., Shleifer, A., Vishny, R., 2002. Investor protection and corporate valuation. Journal of Finance, 57(3), 1147-1170.
- Levinson, H., 1971. Conflicts that plague family business. Harvard Business Review, 49(2), 90-98.
- Martin, H.F., 2001. Is family governance an oxymoron? Family Business Review, 14(2), 91-96.

- Matias, C., Franco M., 2018. Family protocol: how it shapes succession in family firms. Journal of Business Strategy, 41(3), 35-44.
- Mayer, C., 1999. Firm Control. Inaugural Lecture, Said Business School, University of Oxford.
- Miller, D., Le Breton-Miller, I., Lester, R.H., Cannella Jr., A.A., 2007. Are family firms really superior performers? Journal of Corporate Finance, 13, 829-858.
- Minichilli, A., Brogi, M., Calabró, A., 2016. Weathering the storm: family ownership, governance, and performance through the financial and economic crisis. Corporate Governance: An International Review, 24(6), 552-568.
- Molly, V., Laveren, E., Jorissen, A., 2012. Intergenerational differences in family firms: Impact on capital structure and growth behavior. Entrepreneurship Theory and Practice, 36, 703-725.
- Montemerlo, D., Ward, J.L., 2011. The Family Constitution: Agreements to secure and perpetuate your family and your business. Macmillan.
- Morck, R., Wolfenzon, D., Yeung, B., 2005. Corporate Governance, Economic Entrenchment, and Growth. Journal of Economic Literature, 43(3), 655-720.
- Mustakallio, M., Autio, E., Zahra, S.A., 2002. Relational and contractual governance in family firms: Effects on strategic decision making. Family Business Review, 15(3), 205-222.
- Official Journal of the European Union (2016). 15/01/2016. C 13/11. https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:52015IE0722&from=CS
- Patton, M., 1987. How to use Qualitative Methods in Evaluation (No. 4). SAGE.
- Pavón, M., 2018. El Protocolo Familiar. Estructura y Contenido. Ed. Lefebvre.
- Schmid, T., 2013. Control considerations, creditor monitoring, and the capital structure of family firms. Journal of Banking & Finance, 37, 257-272.
- Schulze, W S., Lubatkin, M.H., Dino, R.N., 2003. Toward a theory of agency and altruism in family firms. Journal of Business Venturing, 18, 473-490.
- Schulze, W.S., Lubatkin, M.H., Dino, R.N., 2003a. Exploring the Agency Consequences of Ownership Dispersion among the Directors of Private Family Firms. Academy of Management Journal, 46(2), 179-194.
- Schulze, W.S., Lubatkin, M.H., Dino, R.N., Buchholtz, A.K., 2001. Agency Relationships in Family Firms: Theory and Evidence. Organization Science, 12(2), 99-116.
- Stake, R. E., 1995. The art of case study research. Thousand Oaks, CA: SAGE.
- Steijvers, T., Voordeckers, W., 2009. Private family ownership and the agency costs of debt. Family Business Review, 22, 333-346.
- Suddaby, R., 2006. What grounded theory is not. Academy of Management Journal, 49(4), 633-642.
- Suáre, K.C., Santana-Martín, D.J., 2004. Governance in Spanish family business. International Journal of Entrepreneurial Behaviour & Research, 10(1/2), 141-163.
- Suess, J., 2014. Family governance Literature review and the development of a conceptual model. Journal of Family Business Strategy, 5(2), 138-155.

- Tapies, J., Ceja, L., 2011. Family Protocol: Better to Agree and Put It in Writing. IESE Insight.
- Van den Berghe, L.A.A., Carchon, S., 2003. Agency Relations within the Family Business System: an exploratory approach. Corporate Governance: An International Review, 11(3), 171-179.
- van Essen, M., Strike, V.M., Carney, M., Sapp, S., 2015. The resilient family firm: Stakeholder outcomes and institutional effects. Corporate Governance: An International Review, 23(3), 167-183.
- Villalonga, B., Amit, R., 2006. How do family ownership, control and management affect firm value?. Journal of Financial Economics, 80(2), 385-417.
- Villalonga, B., Amit, R., 2010. Family control of firms and industries. Financial Management, 39(3), 863-904.
- Villalonga, B., Amit, R., Trujillo, M. A., Guzmán, A., 2015. Governance of family firms. Annual Review of Financial Economics, 7, 635-654.
- Yin, R., 2003. Case study research: Design and methods. Newbury Park, CA: SAGE.
- Zahra, S.A., Pearce, J.A., 1989. Boards of Directors and Corporate Financial performance: A Review and Integrative Mode. Journal of Management, 15(2), 291-334.
- Zellweger, T., Kammerlander, N., 2015. Family, wealth and governance: An agency account. Entrepreneurship Theory and Practice, 39(6), 1281-1303.

Table 1. Geographical case distribution

Region	Firms by region	Population with respect to the national average*	Public aid for implementing family constitution
I	2	Lower tertile	Yes
II	1	Lower tertile	No
III	3	Upper tertile	No
IV	2	Upper tertile	Yes

^{*} lower tertile: up to 2.5 million inhabitants; upper tertile: more than 5.5 million inhabitants.

Table 2. Characteristics of the family firms interviewed

Firm	Industry	Age (Years)	Operating Income (€ mill.)	Size*	Latest Generation
A	Energy	over 80	up to 20	small	4
В	Beverage	over 80	up to 20	medium	5
C	Industrial	under 40	more than 40	medium	2
D	Building	between 40 and 80	up to 20	small	2
Е	Insurance	over 80	up to 20	large	3
F	Industrial	between 40 and 80	up to 20	medium	4
G	Fashion	under 40	between 20 and 40	large	2
Н	Retail	over 80	more than 40	large	3

^{*}Number of employees: small, up to 50; medium, 50 to 250; large, more than 250.

 Table 3. Position of the interviewees

Interviewed	Position Generation			
A1	CEO	3		
A2	Managing Director	4		
A3	Manager for SMEs (bank clerk)	external (creditor)		
B1	Vice president	4		
B2	Managing Director (Founder)	5		
В3	Deputy General Manager (bank)	external (creditor)		
C1	Deputy General Manager	2		
C2	CEO & President (Founder)	1		
С3	Control Manager	external		
D1	Managing Director	2		
D2	Founder	1		
D3	Director of Corporate Development	external (in-law)		
E1	CEO	3		
E2	President	3		
E3	CFO	external		
F1	Executive Director	3		
F2	Executive Director in Mexico	4		
F3	CFO	external		
G1	Managing Director	2		
G2	Founder	1		
G3	Consultant (protocols)	external		
H1	President	2		
H2	Managing Director	3		
НЗ	Consultant (protocols)	external		

Table 4. Ownership, Governance and Management data

Firm	A	В	С	D	Е	F	G	Н
Owners (No.)	7	143	6	2	5	6	3	6
Family owners (No.)	7	58	6	2	3	6	3	4
Family ownership (%)	100%	87.4%	100%	100%	62%	100%	100%	75%
Family branches (No.)	2	6	2	2	3	6	3	4
Minority family owners	no	yes	no	no	yes	no	no	no
Family directors (%)	100%	60%	100%	50%	44%	100%	100%	30%
Family directors (No.)	3	3	4	2	4	3	3	3
Non-Family directors (owners) (No.)	0	0	0	0	2	0	0	3
Independent directors (No.)	0	2	0	2	3	0	0	4
Family CEO	yes	yes	yes	yes	yes	yes	yes	no
CEO duality	no	no	yes	no	no	no	no	no

 Table 5. Reasons for implementing the family constitution

Interviewee comments	When	Reasons		
"After the succession, I had to decide who could work here and who couldn't, and that couldn't continue like this" (B1)	Just after succession	Organization		
"Actually, the reasons why we decided to implement a protocol were succession and disorganization" (D1)	Just after succession	Organization		
"The reason we made a protocol was to bring order to the firm" $(F1)$	Just after succession	Organization		
"We drafted a protocol to improve firm planning and resolve a conflict with an in-law" (G1)	Just after succession	Organization/Conflict		
"The reason for the protocol was the organization, leadership and conflict with a family member" (H1)	Just after succession	Organization/Conflict		
"We realized that every firm that wants to manage itself in a family way has to have a Constitution, which is a protocol" (C1)	Just after succession	Organization/Transition		
"The reason they decided to implement a family protocol is because they wanted a smooth, professional and conflict-free transition" (E1)	Just after succession	Transition		
"The interest arises from a family member who had heard of family protocol in a family business association" (A1)	Stability	Institutional influence		

Table 6. Characteristics of the family constitution document

Firm	A	В	С	D	Е	F	G	Н	Average
Preparation (months)	24	12	12	3	6	12	18	12	12.37
Family Owners	7	58	6	2	3	6	3	4	11.12
Signatories	15	12	6	5	21	7	4	5	9.37
Pages	15	30	105	30	18	20	87	108	51.62
Versions	2	11	1	1	3	1	1	2	2.75
Years since implementation	9	25	2	8	19	13	6	11	11.62